Author’s response to reviews

Title: Clinical consequences of untreated dental caries assessed using PUFA Index and its covariates in children residing in orphanages of Pakistan

Authors:

Ramsha Kamran (ramsha_k@outlook.com)

Warda Farooq (wardafarooq_92@outlook.com)

Mehreen Faisal (mehreen_cdm@margalla.edu.pk)

Faisal Jahangir (dr.fjhangir@gmail.com)

Version: 3 Date: 20 Feb 2017

Author’s response to reviews:

We wish to thank the Editors and the reviewers for their most valuable input on our manuscript and for giving us a chance to get our study published in such a well renowned Journal.

Comment 1: In the Declarations section, under Ethics and Consent to participate, please make it clear, as you state in the Methods section, that written informed consent was obtained.

Response: Suggested changes carried out in manuscript In the Declarations section, under the heading ‘Ethics and Consent to participate’ pg.17 line 11.

Comment 2: As your manuscript does not report on any individual person’s data in any form (including individual details, images or videos), consent for publication is not necessary, so please write "Not applicable" in this section.

Response: Done.

Comment 3: Please format the abstract with the following sub-headings: Background, Methods, Results and Conclusions.

Response: Done.
Comment 4: Please format the main body of the manuscript with the following sub-headings: Background, Methods, Results, Discussion and Conclusions.

Response: Done.

Comment 5: Please move the List of abbreviations to after the Conclusions.

Response: Done.